\begin{thebibliography}{10}

\bibitem{bonda98}
A.~Bondavalli, I.~Mura, and I.~Majzik.
\newblock {Automated dependability analysis of {UML} designs}.
\newblock In {\em Proc. of Second IEEE International Symposium on
  Object-Oriented Real-Time Distributed Computing}, 1998.

\bibitem{mrspn}
H.~Choi, V.~Kulkarni, and K.~S. Trivedi.
\newblock ``{Markov} {Regenerative} {Stochastic} {Petri} {Net}''.
\newblock {\em Performance Evaluation}, 20(1--3):337--357, 1994.

\bibitem{Globus:97}
I.~Foster and C.~Kesselman.
\newblock Globus: A metacomputing infrastructure toolkit.
\newblock {\em International Journal of Supercomputer Applications},
  11(2):115--128, 1997.

\bibitem{Vlissides:94}
E.~Gamma, R.~Helm, R.~Johnson, and J.~Vlissides.
\newblock {\em {Design Patterns: Elements of Reusable Object-Oriented
  Software}}.
\newblock Addison-Wesley, Reading, MA, 1995.

\bibitem{Hatcliff:03}
J.~Hatcliff, W.~Deng, M.~Dwyer, G.~Jung, and V.~Prasad.
\newblock {Cadena: An Integrated Development, Analysis, and Verification
  Environment for Component-based Systems}.
\newblock In {\em Proceedings of the 25th International Conference on Software
  Engineering}, Portland, OR, May 2003.

\bibitem{spnp-new}
C.~Hirel, B.~Tuffin, and K.~S. Trivedi.
\newblock ``{SPNP}: {Stochastic} {Petri} {Nets}. {Version} 6.0''.
\newblock {\em Lecture Notes in Computer Science 1786}, 2000.

\bibitem{hoeben00}
F.~Hoeben.
\newblock ``{Using} {UML} models for performance calculation''.
\newblock In {\em Proc. of Workshop on Software and Performance}, 2000.

\bibitem{fspn-1}
G.~Horton, V.~Kulkarni, D.~Nicol, and K.~S. Trivedi.
\newblock ``{Fluid} stochastic {Petri} nets: {Theory}, application and solution
  techniques''.
\newblock {\em Journal of Operations Research}, 405, 1998.

\bibitem{avail-vax}
O.~Ibe, A.~Sathaye, R.~Howe, and K.~S. Trivedi.
\newblock ``{Stochastic} {Petri} net modeling of {VAXCluster} availability''.
\newblock In {\em Proc. of Third International Workshop on Petri Nets and
  Performance Models}, pages 142--151, Kyoto, Japan, 1989.

\bibitem{poll-perf}
O.~Ibe and K.~S. Trivedi.
\newblock ``{Stochastic} {Petri} net models of polling systems''.
\newblock {\em IEEE Journal on Selected Areas in Communications},
  8(9):1649--1657, December 1990.

\bibitem{queue-perf}
O.~Ibe and K.~S. Trivedi.
\newblock ``{Stochastic} {Petri} net analysis of finite--population queueing
  systems''.
\newblock {\em Queueing Systems: Theory and Applications}, 8(2):111--128, 1991.

\bibitem{king99}
P.~King and R.~Pooley.
\newblock ``{Derivation} of {Petri} net performance models from {UML}
  specifications of communication software''.
\newblock In {\em Proc. of XV Performance Engineering Workshop}, 1997.

\bibitem{menasce98}
D.~A. Menasce and H.~A. Gomaa.
\newblock ``{On} a language based method for software performance engineering
  of client/server systems''.
\newblock In {\em Proc. of Workshop on Software and Performance}, 1998.

\bibitem{rel-srn}
J.~Muppala, G.~Ciardo, and K.~S. Trivedi.
\newblock ``{Stochastic} {reward} nets for reliability prediction''.
\newblock {\em Communications in Reliability, Maintainability and
  Serviceability: An International Journal Published by SAE Internationa},
  1(2):9--20, July 1994.

\bibitem{peterson81}
J.~L. Peterson.
\newblock {\em Petri Net Theory and the Modeling of Systems}.
\newblock Prentice-Hall, 1981.

\bibitem{petriu00}
D.~Petriu, C.~Shousha, and A.~Jalnapurkar.
\newblock ``{Architecture}--based performance analysis applied to a
  telecommunication system''.
\newblock {\em IEEE Trans. on Software Engineering}, November 2000.

\bibitem{pooley99}
R.~Pooley and P.~King.
\newblock ``{The} unified modeling language and performance engineering''.
\newblock {\em IEEE Software}, 1999.

\bibitem{evol-spn}
A.~Puliafito, M.~Telek, and K.~S. Trivedi.
\newblock ``{The} evolution of stochastic {Petri} nets''.
\newblock In {\em Proc. of World Congress on Systems Simulation}, pages 3--15,
  Singapore, September 1997.

\bibitem{corba-perf}
S.~Ramani, K.~S. Trivedi, and B.~Dasarathy.
\newblock ``{Performance} analysis of the {CORBA} event service using
  stochastic reward nets''.
\newblock In {\em Proc. of the 19th IEEE Symposium on Reliable Distributed
  Systems}, pages 238--247, October 2000.

\bibitem{Schmidt:01o}
R.~E. Schantz and D.~C. Schmidt.
\newblock {Middleware for Distributed Systems: Evolving the Common Structure
  for Network-centric Applications}.
\newblock In J.~Marciniak and G.~Telecki, editors, {\em Encyclopedia of
  Software Engineering}. Wiley \& Sons, New York, 2002.

\bibitem{Schmidt:02a}
D.~C. Schmidt and S.~D. Huston.
\newblock {\em C++ Network Programming, Volume 2: Systematic Reuse with ACE and
  Frameworks}.
\newblock Addison-Wesley, Reading, Massachusetts, 2002.

\bibitem{Schmidt:00a}
D.~C. Schmidt, M.~Stal, H.~Rohnert, and F.~Buschmann.
\newblock {\em {Pattern-Oriented Software Architecture: Patterns for Concurrent
  and Networked Objects, Volume 2}}.
\newblock Wiley \& Sons, New York, 2000.

\bibitem{VEST:03}
J.~A. Stankovic, R.~Zhu, R.~Poornalingam, C.~Lu, Z.~Yu, M.~Humphrey, and
  B.~Ellis.
\newblock {VEST: An Aspect-based Composition Tool for Real-time Systems}.
\newblock In {\em Proceedings of the IEEE Real-time Applications Symposium},
  Washington, DC, May 2003. IEEE.

\bibitem{man-perf}
H.~Sun, X.~Zang, and K.~S. Trivedi.
\newblock ``{A} stochastic reward net model for performance analysis of
  prioritized {DQDB} {MAN}''.
\newblock {\em Computer Communications, Elsevier Science}, 22(9):858--870, June
  1999.

\bibitem{NS-2}
{The VINT Project}.
\newblock {Network Simulator - NS-2}.
\newblock \url{http://www.isi.edu/nsnam/ns}.

\bibitem{williams98}
L.~G. Williams and C.~U. Smith.
\newblock ``{Performance} evaluation of software architectures''.
\newblock In {\em Proc. of the Workshop on Software and Performance}, Santa Fe,
  NM, 1998.

\end{thebibliography}

%%% $Id: intro.tex,v 1.8 2005/02/15 09:43:36 gokhale Exp $
\section{Introduction}
\label{intro}

\subsection*{Emerging Trends and Challenges}

% showcase salient features of DHPS systems
Large-scale, distributed, high performance software (DHPS) systems
form the basis of numerous scientific grid computing applications.  
DHPS systems comprise many interdependent artifacts, such as
network/bus interconnects, many coordinated local and remote
endsystems, and multiple layers of software. DHPS systems demand
multiple simultaneous quality of service (QoS) properties including
predictability, controllability, and adaptability of operating
characteristics for applications with respect to such features as
time, throughput, accuracy, confidence, security and
synchronization.  All these issues become highly volatile in
large-scale DHPS systems, due to the dynamic interplay of the many
interconnected parts that are often constructed from smaller parts.

Although it is possible in theory to develop these types of complex
DHPS systems from scratch, contemporary economic and organizational
constraints, as well as increasingly complex requirements and
competitive pressures, make it infeasible to do so in
practice. A key enabler in recent successes with DHPS systems has been
{\em QoS-enabled middleware}~\cite{Schmidt:01o}, such as
Globus~\cite{Globus:97}.

% folks I don't think this figure is needed either - Andy
\iffalse
, as shown in Figure~\ref{fig:mw-layers}. 

\begin{figure}[htb]
\begin{center}
\includegraphics[width=2.1in]{../Sigmetrics-2005/eps/mw-layers2.eps}
\bfcaption{Middleware Stack}
\label{fig:mw-layers}
\end{center}
\end{figure}
\fi

Middleware comprises software layers that provide platform\--independent
execution semantics and reusable services that coordinate how
application components are composed and interoperate. The primary
role of middleware is to (1)~functionally bridge the gap between
application programs and the lower-level hardware and software
infrastructure in order to coordinate how parts of applications are
connected and how they interoperate, (2)~enable and simplify the
integration of components developed by multiple technology suppliers,
and (3)~provide common reusable accessibility for functionality and
patterns by factoring out artifacts from applications into reusable
code. 

% our vision
The flexibility and configurability offered by middleware is
manifested in the large number of reusable software building blocks
and configuration options, which can be used to compose and build large
systems end to end. These building blocks embody good design practices
called patterns~\cite{Vlissides:94,Schmidt:00a}. The choice of the
pattern to use is dependent on the context and the consequences of
using the pattern.

\iffalse
% commented out - Andy
Figure~\ref{fig:posa2-plop} depicts a number of such interacting 
patterns. Many patterns may often provide
similar functionality but their suitability is dependent on the specific
operation contexts. For example, the {\em Active Object} and {\em
Leader-follower}~\cite{Schmidt:00a} patterns provide alternate approaches to 
concurrency with each solution having its own advantages and 
disadvantages. Although the Active Object pattern 
is simple to implement, it incurs an
additional performance penalty due to thread context switching and
message queueing. On the other hand, the Leader-Follower does
not incur these drawbacks, but its implementation is very complex and
might lead to race conditions. Developers of
DHPS systems need to make the right choices of patterns to use in their
solutions, such that the end-to-end QoS and functional requirements of
the DHPS systems are met. The choice of the patterns to use are
driven by various factors including the context in which the DHPS
system will be deployed, concurrency and distribution requirements of
the application, and other QoS concerns, such as end-to-end latency
and timeliness requirements for real-time DHPS systems or
throughput for other applications (e.g., a telecommunication call
processing switch). 

% Jeff: The figure below is very hard to read when printed on paper.
\begin{figure}[htb]
\begin{center}
\mbox{\includegraphics[width=3in]{../Sigmetrics-2005/eps/posa2-plop.eps}}
\bfcaption{Middleware Patterns and Pattern Languages}
\label{fig:posa2-plop}
\end{center}
\end{figure}
\fi

Current {\em ad hoc} techniques based on manually choosing the right 
set of building blocks are error-prone and may adversely impact
performance, system costs and schedules, since most errors are caught
very late in the lifecycle of DHPS development. It is desirable to
have the ability to analyze the performance of individual building
blocks and the composed system much earlier in the lifecycle of
DHPS systems, thereby significantly lowering system testing costs as
well as improving correctness of the final developed system. 

To address the challenges of performance evaluation of DHPS systems
design, a systematic performance analysis methodology is necessary. 
This methodology would comprise performance models of the 
individual building blocks and the composition of the building blocks.
The performance models are based upon well-known 
%Jeff: mrm and fspn not in the bibtex
% DONE
%analytical/numerical modeling paradigms~\cite{mrm,evol-spn,mrspn,fspn-1} and 
analytical/numerical modeling paradigms~\cite{evol-spn,mrspn,fspn-1} and 
% Jeff: citation needed below
% Ok, I added something - Andy
simulation techniques~\cite{NS-2}. As a first step towards the development
of such a methodology, this paper presents a model of the 
Reactor pattern~\cite{Schmidt:00a}, which provides the very important 
synchronous demultiplexing and dispatching capabilities in 
DHPS systems. The model is based on the 
Stochastic Reward Net (SRN) modeling paradigm~\cite{evol-spn}. 
We illustrate how the model can be used 
to obtain estimates of key performance metrics 
and for the sensitivity analysis of the Reactor pattern with relative ease. 

\subsection*{Paper Organization}

%Jeff: If anyone agrees with my comments about combining sections 2 and 3,
%then the following would need to be changed.
% DONE _ andy
This paper is organized as follows: 
Section~\ref{sec:background} provides an overview of performance
modeling techniques we use followed by related work; 
Section~\ref{sec:reactor_perf} presents the SRN model of the Reactor 
pattern, and illustrates the use of the model through examples;
and finally Section~\ref{sec:conclusions}
offers concluding remarks and directions for future research. 
